
diff -r 0379bd6f3d21 -r 2555552d2c05 handouts/ho02.tex
--- a/handouts/ho02.tex	Tue Sep 30 02:22:16 2014 +0100
+++ b/handouts/ho02.tex	Tue Sep 30 11:34:21 2014 +0100
@@ -109,19 +109,19 @@
 
 
 
-\subsubsection*{Questions}
+%\subsubsection*{Questions}
 
-Coming back to the question of why I use online banking, but 
-prefer not to e-vote. 
+%Coming back to the question of why I use online banking, but 
+%prefer not to e-vote. 
 
-Why do I use e-polling in lectures?
+%Why do I use e-polling in lectures?
 
-Imagine you have a perfectly secure internet voting system, by
-which I mean nobody can tamper with or steal votes between
-your browser and the central server responsible for vote
-tallying. What can still go wrong with such a perfectly secure
-voting system, which is prevented in traditional elections
-with paper-based ballots?
+%Imagine you have a perfectly secure internet voting system, by
+%which I mean nobody can tamper with or steal votes between
+%your browser and the central server responsible for vote
+%tallying. What can still go wrong with such a perfectly secure
+%voting system, which is prevented in traditional elections
+%with paper-based ballots?
 
 \end{document}
 




